Author’s response to reviews

Title: Characteristics of users and implications for the use of complementary and alternative medicine in Ghanaian cancer patients undergoing radiotherapy and chemotherapy: A cross-sectional study.

Authors:

Joel Yarney (kodwoahen@gmail.com)
Andrew Donkor (andydosty35@yahoo.com)
Lily Yarney (babyaraba95@yahoo.com)
Samuel Y Opoku (syopoky@chs.edu.gh)
Isaac Agyeman-Duah (ignitingike@yahoo.com)
Emmanuel Asampong (easampong@chs.edu.gh)
Alice C Abakah (alicemen_sah@yahoo.com)

Version: 4 Date: 26 December 2012

Author’s response to reviews: see over
RE: 1272114123740132 - Complementary and Alternative Medicine Use in Ghanaian Cancer Patients Undergoing Radiotherapy and Chemotherapy: Characteristics of Users and Implications

Dear Sir,

I would be grateful if you consider the revised version of my paper for publication in your journal.

The paper gives a unique perspective to the subject of complementary and alternative medicine by considering its use in a population with high CAM usage outside of infirmity. It examines concurrent use of CAM and conventional treatment, and the paradigm to disease causation which influences use of CAM.

The changes suggested by the reviewers have been implemented, and is presented as follows:

A title page containing the title, full names of authors, their institutions of affiliation, and email address has been created. An abstract and key words have also been included according to the format specified by the journal.

Author contribution, and information as well as a statement on competing interest have been stated.

References and tables have also been presented according to the format of the journal.

Thank you for the opportunity given to us to present our perspective on the use of CAM.

Yours faithfully,

Joel Yarney